Author’s response to reviews

Title: Health-Related Quality of Life and Influencing Factors among Rural Left-Behind Wives in Liuyang, China

Authors:

Jinyao Yi (jinyaoyi@gmail.com)
Bin Zhong (zhongbin5118@126.com)
Shuqiao Yao (shuqiaoyao@163.com)

Version: 4 Date: 28 April 2014

Author’s response to reviews: see over
April 28, 2014

Dear Dr. Simone Vigod and Ms April Rada,

Thanks for your work on our manuscript entitled “Health-Related Quality of Life and Influencing Factors among Rural Left-Behind Wives in Liuyang, China” (MS:7030712841135311). Following your helpful suggestion, we added a statement on ethical approval for the study in the methods section of the revised manuscript (track changes).
We also ensured that our revised manuscript conforms to the journal style.

*Editorial Requirement:*
*Ethical Approval*

*Please provide your methods section a statement on ethical approval for the study. If ethical approval was not obtained, then please state this together with the reason why approval was not needed.*

Response: Thanks for your carefulness. This study was approved by the ethics committee of Central South University, and all participants provided written informed consent at the time of enrollment. We added this information in Methods section of the revised manuscript, as you will see on page 5.

Thank you again for your attention and consideration.

Looking forward to hearing from you.

Sincerely yours,

Shuqiao Yao, M.D.& Ph.D.
Medical Psychological Institute
Second Xiangya Hospital
Central South University
Changsha, Hunan, 410011
P.R. China
Tel.: +86 731 85292130;
Fax: +86 731 85361328,
E-mail address: shuqiaoyao@163.com